International Journal of Linguistics, Literature, and Culture



REVIEW HISTORY

Paper: "Cultural Models for Negotiating Guyanese Identity as Illustrated in Jan Lowe Shinebourne's Chinese Women and Other Fiction"

Corresponding Author: Abigail Persaud Cheddie

Email: luannapersaud@yahoo.com

Doi: 10.19044/llc.v6no3a1

Peer review:

Reviewer 1: Wafaa Abdel-Kader Mostafa Hussein

Suez University, Egypt

Reviewer 2: Blinded

Published: 30.09.2019

